Author’s response to reviews

Title: Search for Cardiac Calcium Cycling Gene Mutations in Familial Ventricular Arrhythmias Resembling Catecholaminergic Polymorphic Ventricular Tachycardia

Authors:

  Annukka Marjamaa (annukka.marjamaa@helsinki.fi)
  Päivi Laitinen-Forsblom (paivi.forsblom@helsinki.fi)
  Annukka M Lahtinen (annukka.lahtinen@helsinki.fi)
  Matti Viitasalo (matti.viitasalo@hus.fi)
  Lauri Toivonen (lauri.toivonen@hus.fi)
  Kimmo Kontula (kimmo.kontula@hus.fi)
  Heikki Swan (heikki.swan@hus.fi)

Version: 5 Date: 7 February 2009

Author’s response to reviews: see over
RE: Manuscript number 1051311299208940, Formatting Changes

Dear Dr Norton and The BioMed Central Editorial Production Team,

Thank you for your letter of January 28th, 2009, implying that our manuscript entitled “Search for Cardiac Calcium Cycling Gene Mutations in Familial Ventricular Arrhythmias Resembling Catecholaminergic Polymorphic Ventricular Tachycardia” by Marjamaa et al. will be published in the BMC Medical Genetics pending formatting requirements. We have now carefully revised the manuscript according to the comments raised by the The BioMed Central Editorial Production Team.

On behalf of the authors,

Heikki Swan, MD, PhD
Corresponding author

Helsinki University Hospital
Department of Cardiology,
University of Helsinki,
P.O. Box 340
FIN-00290 Helsinki,
Finland.
Tel.: +358 9 471 74562;
Fax: +358 9 471 76652.
E-mail address: heikki.swan@helsinki.fi